Author’s response to reviews

Title: Pilot Parallel Randomised Controlled Trial of Protective Socks against Usual Care to Reduce Skin Tears in High Risk People: 'STOPCUTS'.

Authors:
Roy Powell (r.j.powell@exeter.ac.uk; roy.powell@nhs.net)
Christopher Hayward (christopher.hayward@plymouth.ac.uk)
Caroline Snelgrove (caroline.snelgrove@plymouth.ac.uk)
Kathleen Polverino (k.polverino@nhs.net)
Linda Park (linda.park1@nhs.net)
Rohan Chauhan (rohanchauhan@nhs.net)
Philip Evans (philip.evans@pms.ac.uk)
Rachel Byford (Rachel.Byford@cancer.org.uk)
Carolyn Charman (carolyn.charman@nhs.net)
Christopher Foy (chris.foy@glos.nhs.uk)
Colin Pritchard (gandcpritchard@googlemail.com)
Andrew Kingsley (andrew.kingsley@nhs.net)

Version: 2 Date: 28 Jun 2017

Author’s response to reviews:

Dear Editor,

I am uploading a revised version of the main document in which I have reduced the word count of the Abstract to 350 words as requested. I have also changed the words in the additional files to either 'Figure:' or 'Table:' as appropriate in the cases of Figure 1, Figure 3, Table 1 and Table 7 to be consistent with the other Tables and Figures. I am therefore uploading those 4 files again together with the main document.
I hope that this will be satisfactory.

best wishes

Roy Powell